

Dear Dr.

Subject: Response to Reviewer Comments for Manuscript [Effect of comprehensive intervention model based on drug-psychology-society-skills on medication compliance and cognitive ability of chronic schizophrenia patients]

I hope this message finds you well. I am writing to express my sincere gratitude for the valuable feedback and constructive comments provided by the reviewers regarding our manuscript titled "The Impact of a Comprehensive Intervention Model Based on Medication, Psychological, and Social Skills on Medication Adherence and Cognitive Abilities in Patients with Chronic Schizophrenia." Your insights have greatly helped us to improve the quality and clarity of our work.

We have carefully considered each comment and have made the necessary revisions to address the reviewers' concerns. Below, I provide a detailed account of the changes we have implemented:

Abstract and Purpose Section: We have added a sentence to the abstract and the purpose section emphasizing the broader implications of our study for clinical practice and future research. This highlights the significance of our findings and their potential impact on patient outcomes.

Quantitative Improvements: Specific quantitative improvements observed in the study, such as percentage changes in symptom scores and cognitive ability, have been included in both the abstract and the results section of the main text. This provides a clearer picture of the intervention's impact.

Necessity of a Comprehensive Intervention Model: We have clearly stated why a comprehensive intervention model is necessary for schizophrenia patients, detailing how traditional treatment approaches fall short and what specific gaps our study aims to address. This information has been added to the preamble.

Hypothesis: The hypothesis has been presented explicitly, explaining why the combination of drug, psychological, and social skill interventions is expected to be more effective. This content has been added to the preamble.

Potential Confounders: We have discussed potential confounders and how they were controlled for, including the even distribution of socio-demographic variables between groups. This information has been added to the discussion section.

Integration into Standard Care Practices: We have supplemented the methods section and the corresponding parts of the discussion with details on how clinicians can integrate this comprehensive intervention model into standard care practices for schizophrenia.

Conclusion: The conclusion has been revised to recap the most important findings and their implications for improving schizophrenia management. This reinforces the significance of our study and its contribution to the field.

Detailed Interventions: A more detailed step-by-step description of the interventions used in the study has been provided in the methods section. This includes specific psychological techniques, social skills exercises, and skill training protocols, ensuring reproducibility.

Changes from Baseline to Post-Intervention: We have highlighted changes from baseline to

post-intervention in both groups, clearly showcasing the improvements attributed to the comprehensive intervention.

Literature References: The manuscript has been updated with the necessary DOI and PMID references to ensure completeness and accuracy.

We believe that these revisions have significantly enhanced the manuscript and addressed all the reviewers' concerns. We are confident that our work now better conveys the significant benefits of an integrated treatment approach, ultimately contributing to improved chronic schizophrenia management and patient outcomes.

Thank you once again for your time and consideration. We look forward to your feedback and hope that our revised manuscript will meet the standards of [Journal Name].

Best regards,